Author's response to reviews

Title: Effects of polyphenols and lipids from Pennisetum glaucum grains on T-cell activation: modulation of Ca2+ and ERK1/ERK2 signaling

Authors:
Abdelhafid Nani (nanifleches2003@hotmail.com)
Meriem Belarbi (me.belarbi@hotmail.fr)
Wided Ksouri-Megdiche (ksouriwid@hotmail.fr)
Souleymane Abdoul-Azize (bontasso@yahoo.fr)
Chahid Benammar (chabena62@yahoo.fr)
François Ghiringhelli (francois.ghiringhelli@yahoo.fr)
Aziz Hichami (Aziz.Hichami@u-bourgogne.fr)
Naim A Khan (Naim.Khan@u-bourgogne.fr)

Version: 2
Date: 26 June 2015

Author's response to reviews: see over
Dijon, June 23, 2015

Editor-in-Chief,
BMC complementary and alternative medicine

Subject: Submission of our MS entitled, “Effects of polyphenols and lipids from Pennisetum glaucum grains on T-cell activation: modulation of Ca^{2+} and ERK1/ERK2 signaling” for publication

Dear sir,

Please find our MS entitled as above and submitted today via the web site of BMC complementary and alternative medicine. We have made all the necessary corrections / modifications that have been mentioned by the Editorial mail. We feel so sorry for not complying to the journal style and for mentioning wrongly the name of another journal.

We have previously two research papers in your journal. Our MS is pertinent to the scope of the BMC complementary and alternative medicine as it brings the new information on the implication of polyphenols and lipids, extracted from Pearl Millet. It is very tempting to know that plants can be used for the herbal therapy as they exert no side effects.

We also declare that if accepted, our work will not be published elsewhere in the same form, in English or in any other language. No significant amount of data reported in their manuscript has been published elsewhere or is under consideration for publication elsewhere. All listed authors declare no conflict of interest and have seen and approved the manuscript as submitted.

Our article has not been sanctioned by any Editorial committee and all of the authors have no conflict of interest. Our article is not currently under consideration at any journal other than BMC complementary and alternative medicine.

Naim KHAN
Professeur de Physiologie